Author's response to reviews

Title: Risk factors for multi-drug resistant Acinetobacter baumannii bacteremia in patients with colonization in the intensive care unit

Authors:

Ji Ye Jung (stopyes@yuhs.ac)
Moo Suk Park (pms70@yuhs.ac)
Song Ee Kim (dobie@yuhs.ac)
Byung Hoon Park (serandoc@yuhs.ac)
Ji Young Son (sonjyoung@yuhs.ac)
Eun Young Kim (narae97@yuhs.ac)
Joo Eun Lim (withjueun@yuhs.ac)
Sang Kook Lee (iskhi@yuhs.ac)
Sang Hoon Lee (tearpoem@yuhs.ac)
Kyung Jong Lee (drkyung@yuhs.ac)
Young Ae Kang (mdkang@yuhs.ac)
Se Kyu Kim (sekyukim@yuhs.ac)
Joon Chang (chang@yuhs.ac)
Young Sam Kim (ysamkim@yuhs.ac)

Version: 4 Date: 28 July 2010

Author's response to reviews: see over
July 28th, 2010

Dear editor

RE: MS. No. 1044398363730395

Title: Risk factors for multi-drug resistant Acinetobacter baumannii bacteremia in patients with colonization in the intensive care unit

Thank you very much for acceptance of our manuscript. We appreciate your advice about the revision and are very pleased to inform you that we are ready to resubmit our final revised manuscript taking into account your comments. Thank you again for your kind review.

Sincerely yours,

Young Sam Kim M.D.

Assistant Professor
Division of Pulmonology, Department of Internal Medicine
Yonsei University College of Medicine
250 Seongsanno, Seodaemun-gu
Seoul 120-752, Republic of Korea
Telephone: 82-2-2228-1971, Fax: 82-2-393-6884
E-mail: ysamkim@yuhs.ac
Major revisions (we require the author to make these changes)

Abstract: please ensure that the Abstract is present in the manuscript and matches exactly what is in the submission system. Currently they do not match.
→ We have corrected the abstract in the submission system exactly like the manuscript

Authors' contributions: Please use initials instead of full name.
→ We have corrected Author’s contributions using initials instead of full name.

References: References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example 'text [1].’ Reference 23 is not currently cited in the manuscript. Please ensure that all references are cited in consecutive numerical order once this reference has been included.
→ We have reviewed references cited in the text using consecutive numbers in square brackets. We have also cited Reference 23 in the Discussion on page 15.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Line numbers: Please remove these.
→ We have removed the line numbers.

Short running headline: Please remove this from the title page.
→ We have removed “Short running headline” from the title page.

Email address: please include email addresses for all the authors on the title page, using the following format:
AB: abcd@institution.ac.uk
EF: efgh@generic.co.uk
IJ: IJKL@corporation.com
→ We have included e-mail addresses for all the authors on the title page, using the suggested format.

Text: Please ensure that all text is displayed in black throughout the manuscript.
→ We have changed the all the text in black throughout the manuscript.

References: the reference list should contain up to 30 authors' names for each citation. The term ’et al’ should not be used if there are fewer than 30 authors.
→ We have corrected the References containing up to 30 authors’ names.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
We have corrected few misspelled words. Moreover, in the legend of Figure 1, we have added full words of some abbreviations.